Author's response to reviews

Title: Transfers to hospital in planned home births - a systematic review

Authors:

Ellen Blix (ellen.blix@unn.no)
Merethe H Kumle (merethe.kumle@unn.no)
Hanne Kjærgaard (ellen.blix@unn.no)
Pål Øian (pal.oian@unn.no)
Helena E Lindgren (helena.lindgren@ki.se)

Version: 3
Date: 4 May 2014

Author's response to reviews:

Dr Valerie Smith
Editor-in-Chief
BMC Pregnancy & Childbirth

May 4th, 2014

Dear Dr Smith,

Re: Manuscript reference No. 6919737361162398

Please find attached a revised version of our manuscript “Transfers to hospital in planned home births - a systematic review”, which we would like to resubmit for publication as a systematic review in BMC Pregnancy & Childbirth.

According to your recommendations, we have:
1) Changed the subheading “Implications for future studies” to “Conclusions”
2) Provided a list of abbreviations
3) Added “Acknowledgements

We shall look forward to hearing from you at your earliest convenience.

Yours sincerely,

Ellen Blix
University Hospital of North Norway
N-9038 Tromsø, Norway
E-mail: ellen.blix@unn.no